Control of mechanically stable movement by spinalized frogs

by

Andrew Garmory Richardson

B.S. Biomedical Engineering Case Western Reserve University, 2000

Submitted to the Department of Mechanical Engineering in partial fulfillment of the requirements for the degree of

Master of Science in Mechanical Engineering

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

June 2003

© Massachusetts Institute of Technology 2003. All rights reserved.

Author
Department of Mechanical Engineering
$M_{av} = 2003$
Way 5, 2000
Certified by
Emilio Bizzi
Institute Professor, Professor of Brain and Cognitive Sciences
Thesis Supervisor
Certified by
Jean-Jacques E. Slotine
Professor of Mechanical Engineering and Information Sciences,
Professor of Brain and Cognitive Sciences
Thesis Supervisor
Accepted by
Ain A. Sonin
Professor of Mechanical Engineering
Chairman, Department Committee on Graduate Students

Control of mechanically stable movement by spinalized frogs

by

Andrew Garmory Richardson

Submitted to the Department of Mechanical Engineering on May 9, 2003, in partial fulfillment of the requirements for the degree of Master of Science in Mechanical Engineering

Abstract

Evidence suggests that the isolated vertebrate spinal motor system might use only a few muscle synergies for the production of a range of movements. The evolution of such synergies encoded in the spinal cord could be dictated by mechanical stability requirements for interacting with the environment or by particular performance advantages. Previous work in frogs and cats has shown that the isometric forces measured during movements evoked by intraspinal stimulation converge to a stable equilibrium. In non-isometric conditions, however, there is no guarantee that a similar property of convergence will be observed. We therefore characterized the stability properties of trajectories produced by spinalized frogs. Hindlimb movements in frogs were measured and phasic force perturbations were applied by a Phantom robot (Sensable Tech., Inc) attached at the ankle. EMGs were recorded from 12 hindlimb muscles and used to trigger the perturbations in both hindlimb-to-hindlimb wipes and withdrawals. In both behaviors, we found that the final position of the movements was stable in that the ankle trajectory after perturbation moved to the final position of the unperturbed trajectory. Following deafferentiation, wiping movements showed a similar, although weaker, recovery after perturbation. Thus the stability properties found during isometric conditions also hold in dynamic conditions. These results show that spinal neural systems are able to stabilize goal-directed movements.

Thesis Supervisor: Emilio Bizzi Title: Institute Professor, Professor of Brain and Cognitive Sciences

Thesis Supervisor: Jean-Jacques E. Slotine Title: Professor of Mechanical Engineering and Information Sciences, Professor of Brain and Cognitive Sciences

Acknowledgments

I first wish to thank Dr. Matt Tresch for his tremendous encouragement and contribution to this project. I furthermore am grateful for the advise and commitment of my advisors, Professor Emilio Bizzi and Professor Jean-Jacques Slotine.

Special thanks go to Russ Tedrake for many helpful discussions and Margo Cantor for her assistance in this work. Finally, I would like to thank Maura Cahill, Kurt Weaver, and my family for all their support.

Funding for this project was provided by the National Institutes of Health and the Whitaker Foundation.

Contents

1	Introduction				
2	Methods				
	2.1	Surgica	I preparation	13	
	2.2	Experi	mental setup and data collection	15	
	2.3	Experimental protocol		16	
	2.4	Data a	nalysis	18	
		2.4.1	Kinematics	18	
		2.4.2	EMGs	20	
3	Behavioral characteristics				
	3.1	Hindlin	nb-to-hindlimb wiping movements	21	
		3.1.1	Behavioral variability	23	
		3.1.2	Effects of deafferentation	25	
	3.2	Hindlin	nb withdrawal movements	26	
4	Movement stability properties				
	4.1 Perturbation characteristics			29	
		4.1.1	Applied force perturbation	29	
		4.1.2	Perturbed position	31	
		4.1.3	Perturbed velocity	34	
	4.2	2 Final position stability			
		4.2.1	Hindlimb wipe	36	

		4.2.2	Hindlimb withdrawal	37		
	4.3 Trajectory stability					
		4.3.1	Contraction analysis	38		
-	<u>a</u> , 1	•1•		40		
5 Stability control strategy						
	5.1	Kinem	natic evidence from deafferented frogs	43		
	5.2	EMG	evidence from afferented frogs	46		
6	Discussion and conclusion					

List of Figures

2-1	Schematic of experimental setup and spinal frog behaviors	15
2-2	Coordinate frames for kinematics analysis	19
3-1	Example hindlimb wipe motor pattern and kinematics $\ldots \ldots \ldots$	22
3-2	Mean wiping movements and the biomechanics underlying variability	24
3-3	Afferent influence on wipe motor pattern and kinematics \ldots .	25
3-4	Example hindlimb with drawal motor pattern and kinematics $\ . \ . \ .$	26
4-1	Example hindlimb wipe perturbation trials	30
4-2	Initial displacements for hindlimb wipe movements $\ldots \ldots \ldots$	32
4-3	Initial displacements for hindlimb with drawal movements $\ . \ . \ . \ .$	33
4-4	Perturbed velocities for hindlimb wipe and withdrawal movements	35
4-5	Final position summary for hindlimb wipe movements	36
4-6	Final position summary for hindlimb with drawal movements $\ . \ . \ .$	37
4-7	Contraction analysis in ankle coordinates	39
4-8	Contraction analysis in alternate coordinates	41
5-1	Perturbation and stability summary for deafferented movements	44
5-2	Example perturbed ankle paths from deafferented frogs \hdots	45
5-3	Relationship between initial and final displacements for wiping move-	
	ments	46
5-4	Mean EMGs for two afferented wiping frogs	48

Chapter 1

Introduction

Movement stability is critical for successfully maneuvering about and interacting with the physical environment. The unexpected, or improperly anticipated, presence or absence of a force during motor behaviors can lead to ineffective movement. The huge variety of circumstances under which perturbations can occur, often requiring a quick compensatory response, favors the large majority of perturbations to be handled by a rapid stabilizer rather than iteratively through neural learning systems. The neuromuscular system generally acts as the stabilizer of animal behavior, although geometrical and mechanical properties of the skeleton may provide compensation in certain situations [25]. Neuromuscular control can act at several levels to provide stability. First, the neural activation-dependent viscoelastic properties of muscle potentially provide passive stabilization at the peripheral level. Second, afferent feedback can drive active compensatory action at the spinal and supraspinal levels. However, nerve transmission delays significantly complicate a stability control strategy relying on supraspinal feedback loops, particularly for high bandwidth movements. In light of this, one might expect a significant reliance on the stabilizing action of peripheral viscoelastic properties and spinal reflexes. In the experiments described in this thesis, we investigate this hypothesis by studying the ability of a spinalized animal, the frog. to recovery from transient perturbations. Subsequently, mechanisms controlling any recovery were explored. We show that movement produced by the frog spinal motor system is stable about behavioral goals and that the stabilization is substantially provided by the mechanical impedance derived from intrinsic muscle properties.

Control theories based on peripheral and spinal stabilizing action

The idea that peripheral properties can stabilize movement has been a central supposition of several major theories of motor control. Prominent among them is the equilibrium point hypothesis. Originally proposed by Feldman [9], the equilibrium point hypothesis is based on the observation that muscles are better described as spring and damper elements than state-independent force generators. Furthermore, the relationship of muscle length and velocity to force can be modified by motorneuron input [20]. Short-latency spinal reflexes can also add to the total spring-like behavior of the neuromuscular system [28], and can likewise be modified by the nervous system. Thus by conceptually simplifying the motor periphery to a set of tunable springs, movement from one point to another may be controlled by changing the rest length of the springs such that the new equilibrium point they define coincides with the desired movement goal. In this way, stabilizing properties of muscles could be used to actually produce movements in essentially the same fashion as a proportionalderivative servo controller for artificial control systems. One of the key benefits of this biological servo control strategy is that, as with its artificial analog, it precludes the need for an inverse dynamics computation and maintenance of a detailed internal model of the motor periphery [30]. In other words, movements can also be implemented by the nervous system in terms of kinematic variables (i.e. muscle lengths) and the dynamics of movement naturally emerge from the interplay between stabilizing neuromuscular forces and skeletal and environmental loads. Elaborations on this hypothesis have been made, in light of data indicating more than just the final position of movement is controlled [4], such that a time series of equilibrium points, known as virtual trajectories, may be specified by the neuromuscular system [16]. Also, there has been considerable debate regarding whether desired equilibrium positions are specified by α -motorneuron input, setting intrinsic muscle stiffness (α model [6]), or by γ -motorneuron input, setting spinal reflex threshold (λ model [10]), with the answer likely being some combination of the two [30].

A related theory to the equilibrium hypothesis is impedance control, proposed by Hogan [18]. Extending the concepts of the equilibrium point hypothesis, which utilizes the position-dependent spring-like behavior of the neuromuscular system, this theory unites into a common framework the position-, velocity-, and acceleration-dependent components of mechanical impedance. The nervous system theoretically has a measure of control over both the amplitude and direction of each of these components in the multijoint limb [17]. In particular, muscular redundancy facilitates control of apparent limb stiffness and viscosity while kinematic redundancy of the skeleton is necessary for control over apparent limb inertia. The hypothesis is that the nervous system actively controls each component of mechanical impedance for increased control of limb stabilization. Advantages of impedance control for multijoint limbs may be realized in tasks where stability is required in certain directions but may hinder performance in others. Experimental evidence suggests that, in some tasks, modulation of stiffness [7] and viscosity [26] orientation can occur subconsciously. However the ability to voluntarily control all the mechanical impedance components of a limb has not yet been shown [36].

One primary assumption made by both the equilibrium point hypothesis and the impedance control hypothesis is that forces produced by muscle viscoelasticity and spinal reflex action can be made sufficiently large to move the inertial load of the limb and compensate for environmental perturbations. Thus we turn to evidence in the literature exploring the stability properties of the neuromuscular system.

Experimental evidence for peripheral and spinal stabilizing action

To address concerns regarding the magnitude of stabilizing forces produced by intrinsic and reflex mechanisms, a number of studies have attempted to quantify some or all of the components of mechanical impedance during posture [31] [8] [42] and movement [3] [13] in humans. Impedance was generally determined by applying a controlled force or displacement perturbation, measuring restoring forces, and fitting the mechanical response to a parametric or nonparametric [35] model. Interpretation of the significance of impedance values varied across studies. Generally, stiffness during movement was found to differ from posture, but the resulting impedance values were dependent on a simplified model of the neuromusculoskeletal system. A modelindependent study simply inferred from physical principles that the restoring force direction to a constraint perturbation had a large position-dependence, thus providing spring-like stability [43]. Two recent human studies have taken yet a different approach to determining the influence of the spring-like nature of the neuromuscular system on movement stability. Both employ a kinematic, rather than impedance or restoring force, characterization of stability, while ensuring subjects did not make voluntary corrective movements following the perturbation. The first study applied phasic force perturbations during small amplitude, point-to-point movements and found that the final position was not achieved by the subjects [37]. The second study loaded nominal wrist movements and then perturbed the movements by unloading the movement [14]. As with the previous experiment, the final position was not achieved in perturbed trials indicating intrinsic and reflexive actions may not be sufficient to compensate for perturbations, casting doubt on their role as servo controllers.

In addition to the human studies, a separate line of experimental investigation has been pursued in the frog and cat that relates to the nature of stabilizing forces organized by spinal motor circuits. In these experiments, intraspinal microstimulation was applied and isometric forces of the hindlimb were recorded with the limb in varying configurations [12] [27]. For a given spinal stimulation site, ankle isometric forces at each limb configuration pointed toward a particular point in the workspace, thus specifying a convergent force pattern. This arrangement of force directions would be expected to stabilize the limb about a particular configuration. A relatively small number of different patterns were found to exist [5]. Finally, when two spinal sites corresponding to two different patterns where stimulated simultaneously, the resulting force pattern was a vector sum of the force pattern associated with stimulating each site independently [32]. The linear combination of a small number of force patterns was interpreted as a possible mechanism for simplifying control, by reducing the excess degrees of freedom provided by the redundant musculature. The stability properties of the force patterns may suggest that some structure embedded in the spinal cord can coordinate specific combinations of intrinsic muscle properties and reflex loops into stable functional units. These stable units, or primitives, could perhaps be used in a servo control strategy based on ankle position rather than individual muscle lengths. However, data from the human experiments suggests that stability seen in posture does not necessarily translate into stable movement. Indeed, studies have found instabilities in movement deprived of any voluntary corrections [37] [14]. Thus the stabilizing ability of these spinal primitives seen in isometric conditions many not hold in movement of the frog or cat hindlimb.

The experiments described here extend this previous work in the frog, assessing the dynamic stability of movements. The results of this study are both specifically applicable to frog and cat intraspinal stimulation experiments, as well as generally to the ability of peripheral and spinal mechanisms to stabilize movement in the absence of supraspinal control.

Outline of chapters

Chapter 2 describes the methods employed in carrying out the frog experiments, including surgery, experimental setup, data collection and analysis.

Chapter 3 provides a brief description of the nominal behaviors elicited from the spinalized frog, discussing the variability observed across and within frogs as well as the effects of deafferentation on the basic motor pattern.

Chapter 4 gives the main stability results. The commanded perturbations are described and their effect on the hindlimb kinematics is quantified. Stability about the final limb position is determined. A tool from nonlinear systems analysis, contraction analysis, is applied to determine stability of the full trajectories.

Chapter 5 presents an analysis of the controller stabilizing the system. Intrinsic and reflex mechanisms are differentiated using an EMG analysis and results from deafferented frogs.

Chapter 6 discusses the results in the context of the background presented in this chapter.

Chapter 2

Methods

2.1 Surgical preparation

All procedures were approved by the Committee for Animal Care at MIT. 40 adult bullfrogs (*Rana Catesbiana*) were used in these experiments. Animals were anesthetized with a combination of tricaine (5% solution, 0.5-0.7 ml) and ice anesthesia. The skin overlying the skull was opened, and the muscle tissue overlying the foramen magnum between the first vertebra was removed. The foramen magnum was opened and the skull overlying the tectum and brainstem was removed. The dura was opened and the exposed spinal cord aspirated at the caudal margin of the fourth ventricle, completely separating the spinal cord from the brainstem. All neural tissue rostral to the brainstem was then removed by aspiration and the remaining cranial cavity was packed with gelfoam to minimize bleeding. Gelfoam was also placed over the exposed spinal cord and brainstem and the wound was close using staple sutures.

Animals were then implanted with bipolar electromyographical (EMG) electrodes. Electrodes were constructed from braided stainless steel wires, insulated with Teflon. (7 strands, 0.023 mm). Pairs of wires were knotted together at one end and the knot covered with modeling wax to insulate the two cut ends, forming a wax ball at the end of the electrode pair. The Teflon insulation covering each wire was then stripped for a length of ≈ 1 mm, separated from the wax ball by a distance of $\approx 1-2$ mm. The wires were then inserted in the muscle, so that the stripped regions of the electrode pair were located inside the muscle belly with the pair secured by the wax ball. The orientation of the electrode pair was parallel to the orientation of the muscle fibers. The following muscles were implanted: semitendinosus (ST), sartorius (SA), rectus internus (RI), adductor magnus (AM), vastus internus (VI), semimembranosus (SM), vastus externus (VE), biceps femoris (BF), iliopsoas (IP), rectus anterior (RA), tibialis anterior (TA), and gastrocnemius (GA). EMG electrodes were tunneled subcutaneously to an exit point on the back and attached to a connector strip.

The dorsal tibia just proximal to the ankle was then exposed. Two bone screws were placed in the tibia, dental cement poured around them, and a vertically oriented threaded attachment placed in the cement. The attachment was used to couple the frog hindlimb to the robot during experiments (see below).

In some frogs (8) we also performed a deafferentation. For this procedure, we exposed the spinal cord by dorsal laminectomy of the 6th vertebra. The dura overlying the spinal cord was opened by bipolar electric cautery. In initial experiments, the 7-9th lumbar spinal dorsal roots were identified by the lack of motor response observed following electrical stimulation with silver hook electrodes placed around each nerve. In later experiments, these dorsal roots were identified visually. After identifying the dorsal roots and bathing the exposed spinal cord in tricaine, the dorsal roots were cut. We observed no motor response on cutting the roots, even though animals were usually reflexive prior to bathing the exposed spinal cord in anesthetic, suggesting a lack of injury discharge following sectioning of dorsal roots. Gelfoam was placed on the exposed spinal cord, and the wound was closed in layers, suturing the muscles and then the skin together. Completeness of the deafferentation was confirmed the following day by the absence of all motor responses evoked from scratching sites on the deafferented hindlimb.

Following these procedures, animals were placed in a refrigerator and allowed to recover overnight.

2.2 Experimental setup and data collection

The following day, animals were placed on a stand and secured by a clamp placed around their pelvis. The hindlimb was attached to a Phantom 1.0 robot (Sensable Technologies) using the threaded attachment cemented to the tibia as the point of attachment (Fig. 2-1A). This attachment point between the robot and the animal was adjusted so that the frogs ankle was placed in the plane of the hip. The movement of the hindlimb was restricted to this plane by the robot. These restrictions allowed the hindlimb configuration to be determined uniquely from the endpoint of the robot.



Figure 2-1: Schematic of experimental setup and spinal frog behaviors

A pair of stimulating electrodes was placed on the skin of the foot to evoke either hindlimb-hindlimb wipes or withdrawal behaviors in the limb. Hindlimb wiping in the limb attached to the robot was evoked by placing the stimulating electrodes on the limb contralateral to the attached limb (Fig. 2-1B). Withdrawal was evoked by placing the electrodes on the limb attached to the robot (Fig. 2-1C).

In hindlimb wipes, electrical stimulation evoked a movement of both the stimulated hindlimb and the hindlimb attached to the robot toward the midline (see Chapter 3). To prevent the two hindlimbs from colliding and affecting the measured kinematics, we placed a barrier near the animal's midline. This barrier guaranteed that on repeated wipes, the trajectory of the measured hindlimb was not affected by the contralateral leg.

Activity on EMG electrode pairs was differentially amplified (x1000) and filtered (10-10000 Hz passband, A-M systems) and sampled at 1000 Hz using Labview software (National Instruments). Kinematic data and the forces applied to the hindlimb attached to the robot were measured on a computer running custom software written using the Ghost software package (Sensable Technologies). Kinematic and force data were collected at 1000 Hz. All data was saved for offline analyses.

2.3 Experimental protocol

We examined the trajectories and EMG patterns produced in hindlimb wiping and in withdrawal behaviors. In all cases of hindlimb wiping, the trajectory and EMG activity of only the unstimulated, wiping, limb was measured. For both behaviors, we examined trajectories in unperturbed trials and in trials in which a phasic perturbation was applied.

For unperturbed trials, a behavior was evoked by electrical stimulation applied to either the ipsilateral or contralateral foot. The trigger for stimulation onset was given by the computer running the robot software. This trigger was also used to initiate data collection on the EMG computer. In separate experiments using an external clock sent to both computers, we found that there was constant 3 ms delay between the trigger sent by the Phantom and the start of EMG data collection, with no variability in this delay over repeated triggers. This delay was subtracted in all data sets to align EMG and kinematic data. Electrical stimulation consisted of trains of biphasic stimulus pulses (train duration: 600-1700 ms, train frequency: 25-35 Hz, pulse duration: 1-2 ms, pulse amplitude: 1-1.8 mA). Intervals of 2-5 minutes were allowed between repeated stimulation trains, so as to minimize habituation or potentiation of behaviors. Kinematic and EMG data were recorded for a period of 4 s following the onset of the electrical stimulation.

For perturbed trajectories in both wipes and withdrawals, the onset of the perturbation was triggered from the observed EMG pattern. This trigger based on EMG activity was used rather than one based on a fixed time following stimulation onset in order to ensure that the perturbation was applied at a similar time in the evoked motor behavior. Perturbations were applied early in the evoked motor pattern, usually within the first 100 ms of the EMG activity. Several observations suggest that the sensitivity of frog hindlimb movements to perturbations is especially high during the early phases of movement production [34]. Perturbations applied early should therefore have the highest probability of evoking measurable compensations in observed behaviors. Integrated EMG from the earliest activated muscle (ST, BF, or IP; determined for each frog individually) was measured online in 5 ms bins and accumulated until it crossed a threshold. When this threshold was crossed, a perturbation was applied to the frog hindlimb. The threshold was chosen for each frog so that the perturbation was triggered within the first 100 ms of observed EMG (see Figure 4-1A) but was insensitive to baseline noise. Perturbations were applied in two different directions, defined with respect to the instantaneous velocity of the ankle. Clockwise (CK) perturbations were applied at an angle of +90 to +135 degrees while counterclockwise (CCK) perturbations were applied at an angle of -90 to -135 degrees. CK perturbation trials, CCK perturbation trials, and unperturbed trials (O), make up the three perturbation groups referred to in the data analysis. The perturbation command consisted of a square pulse, amplitude 0.35 to 1.50 N and duration 25 to 75 ms. We confirmed the peak amplitude and duration of the perturbation in separate experiments using a force transducer attached to the robot. Care was taken so that the applied perturbation did not drive the limb into the boundaries of the frogs workspace, which could be observed by a clear and immediate return of the hindlimb to the unperturbed trajectory following the perturbation. Any such trials were excluded from further analysis.

2.4 Data analysis

All analyses of kinematic and EMG data were performed using MATLAB software (MathWorks).

2.4.1 Kinematics

The perturbed kinematics of the hindlimb wipe and withdrawal behaviors were used to determine stability. To prepare the kinematics for analysis, the goal-directed portion of the hindlimb trajectories was determined using criteria similar to that used in [38]. For hindlimb wipes, the portion of the observed trajectory up until collision with the midline barrier was identified. For withdrawal reflexes, the portion of the observed trajectory up until collision with the body of the frog was identified. Besides observing the trajectory reaching the boundaries of the hindlimb workspace, these collision points could be seen in a sharp transient of the hindlimb velocity profile. These collision points were considered to be the end of the movement for the purposes of the subsequent analyses.

Prior to the stability analyses, the degree to which the perturbation caused a change in hindlimb kinematics was quantified. Perturbed position was measured relative to the unperturbed mean path. To calculate this path, the path of each individual unperturbed trajectory was spatially resampled, using spline interpolation, such that there was an equal distance between each point along the path from initial to final position. The resampled paths were then averaged, point by point, to obtain the mean unperturbed path. The variance and 95% confidence intervals of this mean path were calculated from these same distributions of resampled trajectories. The initial displacement due to the perturbation was taken as the maximum displacement from the mean unperturbed path, calculated along directions perpendicular to the path, in the 100 ms following perturbation offset. This time window included the pe-

riod of time during which the hindlimb was still moving away from the unperturbed path after the initial acceleration caused by the perturbation. For reference, this distance was also calculated for unperturbed trials, where time of perturbation offset was determined using the same EMG threshold criterion of when a perturbation would have been applied. Initial displacements were also analyzed in the joint space of the frog hindlimb. For this analysis, we used link lengths measured from x-rays taken post mortem and measurement of the frog's hip position to transform the ankle positions of the hindlimb to joint coordinates (as defined in Figure 2-2). In addition



Figure 2-2: Coordinate frames for kinematics analysis

to perturbed position, the effect of the perturbation on limb velocity was assessed in both joint space and muscle space. For the latter, we used a detailed biomechanical model of the frog hindlimb developed by Bill Kargo and Larry Rome [23]. Using the configuration-dependent moment arms of ST and BF provided by this model, along with the joint velocities calculated from the observed trajectories, we obtained an estimate of the velocity of muscle lengthening or shortening caused by the applied perturbations.

Two stability analyses were performed using the kinematic data. The first compared the final position of unperturbed trials to that of perturbed trials to test stability about the final position. The second tested the tendency for each pair of CK and CCK perturbation trajectories to converge towards each other, using contraction analysis [29].

2.4.2 EMGs

EMGs were analyzed primarily for the purpose of determining the role of spinal reflexes in compensating for the phasic perturbations. All EMGs were rectified and then digitally filtered (acausal 5th order low pass Butterworth filter, 20 Hz nominal cutoff). Onsets and offsets of muscle activations were determined using an automatic detection routine. These values were verified visually for each trial and adjusted when necessary. For a qualitative analysis of EMG alterations following the perturbation, mean EMGs were computed for each perturbation group of each frog after aligning the EMGs of each trial on perturbation onset. For the unperturbed group (O), trials were aligned on when the perturbation would have occurred, using the same EMG threshold criterion mentioned above. For a statistical analysis of EMG alterations following the perturbation, we looked at perturbation group differences in three parameters: latencies between onset of different muscles, duration of individual muscles, and magnitude of individual muscles. The duration of EMG response was taken as the time from onset to offset of each muscle. The magnitude of each muscle activation in a trial was calculated as the total integrated EMG for that muscle from perturbation onset to time of final position as defined by the kinematics. Of the six frogs comprising the withdrawal movement data set, two had poor ST or BF EMGs and thus were excluded from all EMG analyses. Each test was performed using a non-parametric Kruskal-Wallis ANOVA.

Chapter 3

Behavioral characteristics

This chapter briefly describes the typical motor patterns and kinematics of nominal, i.e. unperturbed, hindlimb wipe and withdrawal behaviors. Variability in the nominal motor patterns is analyzed to identify some features that can be modulated by the spinal circuits involved in these behaviors. The role of afferent feedback in modulating these features is also described. Changes in the relevant motor pattern parameters identified in this chapter are later considered, in Chapter 5, as possible afferent-driven stabilizing mechanisms.

3.1 Hindlimb-to-hindlimb wiping movements

The primary spinal frog behavior used in this thesis is the hindlimb-to-hindlimb wipe (referred to henceforth as hindlimb wipe or wipe). Previous investigations have used this behavior as a tool for studying sensory to motor mappings and controlled movement variables by vertebrate spinal circuits [11] [38]. The behavioral goal is to use the ankle of one hindlimb to wipe off an irritating stimulus on the other hindlimb (Fig. 2-1B). A particular advantage of this behavior is that it demands relatively precise multijoint movement to remove the stimulus, presumably requiring more sophisticated neural control than ballistic movements such as kicking or withdrawal.

A typical example of the motor program, as determined from EMG activity, and kinematics of a hindlimb wipe is shown in Figure 3-1. The degree of sophistication



Figure 3-1: Example hindlimb wipe motor pattern and kinematics

of the movement can perhaps be seen most clearly in the staggered, rather than synchronous, EMG activity of hindlimb muscles (Fig. 3-1A). The muscle activity can be divided into three phases, as roughly demarcated by the vertical lines: an initial phase involving predominantly knee flexors (IP, ST, BF), a second phase involving predominantly hip extensors (RI, SM), and a final phase involving a knee extensor (VE) as well some muscles acting at the ankle joint (GA, RA, TA). The putative biomechanical action of the muscles just described is in agreement with the joint kinematics (Fig. 3-1B). From the initial position, indicated by the square, the first part of the movement is dominated by knee flexion, followed by hip extension, and ends with knee extension (refer to Figure 2-2 for sign convention of joint angles). In ankle coordinates, the first two phases of muscle activity correspond to a rostral-medial movement toward the midline to place the two limbs in contact with each other (the RC and ML axes arrows point rostral and medial, respectively, as indicated by orientation of the inset frog). Thus the first two phases are referred to collectively as the placing phase in this thesis. This is followed by a caudal extension of the limb contralateral to the stimulus to wipe off the irritating stimulus on the other limb, referred to as the whisking phase [11]. The velocity profiles in joint and ankle coordinates are fairly monophasic for the placing phase of the movement (Fig. 3-1C). A sudden decrease in velocity occurs upon contact with midline barrier (see Section 2.2), followed by a distinct velocity peak for the whisking phase. These characteristics of the wiping behavior are essentially the same as those reported in previous studies [21] [22].

For the purpose of evaluating stability properties, the placing phase of the hindlimb wipe movement is of most interest. This phase has a well defined final position, namely the point approximately along the midline of the frog where the wiping limb makes contact with the midline barrier, making it similar to point-to-point movement tasks often used in evaluating stability properties in humans [37] [43]. One difference, however, is that the velocity is generally nonzero at the final position of the placing phase. To isolate the placing phase, as mentioned in Section 2.4.1, kinematic records were truncated at the point of midline contact (indicated by an arrow in Figure 3-1).

3.1.1 Behavioral variability

Variability in the placing phase of the wipe is largely manifest in the rostrocaudal position along the midline at which the contralateral limb makes contact. Mean unperturbed ankle paths and 95% confidence intervals on the mean (assuming a Student's T distribution) are shown for five frogs in Figure 3-2A. Across different frogs, or different recording sessions in the same frog (16 frogs, 18 sessions for the combined set of afferented and deafferented wiping behaviors), the mean final rostrocaudal position had a range of 48.8 mm. However within a recording session of a given frog, the unperturbed paths were fairly consistent with a mean range of 9.9 mm.

We examined whether the observed variability in the final rostrocaudal position



Figure 3-2: Mean wiping movements and the biomechanics underlying variability

could be related to changes in the EMG patterns. Nine relationships were tested: RC final position as a function of five muscle magnitudes (IP, ST, BF, RI, and SM; magnitude defined as in Section 2.4.2) and as a function of four onset latencies between muscles (ST-RI, ST-SM, BF-RI, and BF-SM). These EMG parameters were chosen due to their relevance in the behavior as identified above and in previous studies [21] [22]. The final rostrocaudal position variability between the 18 hindlimb wipe data sets could in large part be accounted for by the latency between knee flexor and hip extensor muscle activity of the placing phase (Fig. 3-2B). In particular, the latency between onset of ST and onset of RI was significantly correlated to the mean final rostrocaudal positions (p < 0.005) and accounted for 40% of the position variance. Shorter latencies, i.e. longer RI contributions to the movement, correspond to caudal final positions while longer latencies, i.e. less RI contribution, lead to more rostral positions. This relationship is consistent with the biomechanics, as longer latencies allow knee flexor torques to drive most of the movement to the midline, bringing the ankle more rostral. The other eight EMG parameters were not significantly correlated with final RC position in these data sets.

3.1.2 Effects of deafferentation

Deafferentation of the spinalized frogs provided a means of separating the reflex and intrinsic, i.e. muscle viscoelastic, contributions to movement stabilization. However we also found that deafferentian modifies the nominal behavior, as described in [21]. Of the eight frogs that were deafferented, two underwent testing both in the afferent and deafferented conditions. The knee flexors (ST,BF) had a decreased amplitude



Figure 3-3: Afferent influence on wipe motor pattern and kinematics

and knee flexor to hip extensor latency was reduced in the deafferented condition (Fig.

3-3A, EMGs have been rectified, filtered, aligned, and averaged across trials). Both of these observations are consistent with a more caudal mean final ankle position in the deafferented animal (Fig. 3-3B). These observations are also consistent with a recent quantitative analysis of afferent influences on hindlimb wipe behaviors of spinalized frogs [21].

3.2 Hindlimb withdrawal movements

In addition to the 16 spinal frogs used to study hindlimb wiping movements, seven frogs were used to test the stability properties of hindlimb withdrawal movements. The goal of hindlimb withdrawal behaviors is to quickly move the hindlimb away from



Figure 3-4: Example hindlimb withdrawal motor pattern and kinematics

an aversive stimulus (Fig. 2-1C). Withdrawals are more of a ballistic movement than hindlimb wipes, involving synchronous activity of the flexor muscles: IP, ST, and BF (Fig. 3-4A). These muscles both cause flexion of the knee and hip, moving the ankle in a rostral-medial direction until contact is made with the body (Fig. 3-4B). Like the placing phase of the hindlimb wipe, velocity profiles are monophasic up to the point of contact (Fig. 3-4C). For the stability analysis, the final position for this behavior is defined as the point at which the ankle first touches the body (indicated by an arrow in Figure 3-4).

The characteristics of the nominal wipe and withdrawal behaviors observed in our experiments are consistent with those found in previous studies [21] [22] [38]. We found that variability in the wipe movements, as parameterized by final rostrocaudal position, can be significantly correlated with the latency between the onset of ST and RI activity. Furthermore, this latency is modulated by afferent feedback, as determined from the changes in motor patterns following deafferentation. Thus we can hypothesize that changes in ST-RI latency may contribute to perturbation compensation in the wipe, although the other afferent-driven contributions will be tested in Chapter 5 as well.

Chapter 4

Movement stability properties

The ability of the spinal frog to compensate for perturbations during hindlimb wipe and withdrawal movements is examined in this chapter. First, the phasic perturbations that were applied to the hindlimb by the Phantom robot are characterized in terms of the extent to which they changed the kinematics of the limb relative to unperturbed trials. This characterization is used to indicate whether the perturbed kinematics exceeded the variability of the movement, a necessary condition for inferring any subsequent compensation. The analysis can also suggest whether these kinematic changes could be sensed by the frog proprioceptive system, based on known sensitivity of frog muscle spindles. This is an important consideration for the next chapter, which explores the stabilizing strategy employed by the spinal frog. Second, stability is assessed in terms of both final position and trajectory. The assessment of the latter utilizes contraction analysis, a tool from nonlinear systems theory that tests for a type of stability particularly amenable for a system to be embedded in or a result of a distributed control architecture.

4.1 Perturbation characteristics

4.1.1 Applied force perturbation

The onset of phasic perturbations of the hindlimb movements was triggered when a threshold IP, ST, or BF EMG activity was exceeded. As shown in the last chapter, these muscles generally were the first activated in both wipe and withdrawal movements. Since the stability analysis employed in this thesis is based on the kinematics, rather than restoring forces or impedances, timing the perturbations relative to these muscles allowed the perturbations to occur sufficiently early in the movement to determine whether the perturbed kinematics tended toward the unperturbed kinematics prior to midline (wipe) or body (withdrawal) contact. The EMG-based perturbation timing also ensured that muscles were active at the time of the perturbation, thus increasing the likelihood that muscle spindles would sense the perturbation. This rationale is valid for the frog since it does not have independent alpha and gamma motor neuron drive, unlike higher vertebrates, but rather a beta system which couples efferent activity to extrafusal and intrafusal muscle fibers. Finally, EMG-based perturbation timing improved the consistency of the perturbation, implementing the perturbation at approximately the same point in the motor program from trial to trial. An example of this consistency is shown in Figure 4-1A. The EMGs for two perturbation trials of hindlimb wiping movements are shown, both aligned to stimulation onset. The vertical bar indicates the perturbation onset, as triggered by IP activity, and duration. The second perturbation trial shown had a slightly longer latency from stimulus onset to motor response than the first, but due to the EMGbased triggering the perturbations occurred at the same point in the motor pattern.

The direction of the applied perturbation was either 90-135 degrees clockwise (CK) or counterclockwise (CCK) to the ankle movement direction at the time of perturbation onset. These directions permitted the spatial characterization of the perturbation effect, i.e. displacement from the mean unperturbed path, rather than the temporal measures a purely assistive or resistive perturbation would likely require. Example CK and CCK perturbed paths, in ankle and joint coordinates, are



Figure 4-1: Example hindlimb wipe perturbation trials

shown in Figure 4-1B with initial position indicated by a square. The portion of the path where the force perturbation was applied is indicated in white. The grey region indicates the 95% confidence interval for the unperturbed mean path. The maximum displacements from the unperturbed mean path in these examples are substantial, approximately one-third to one-half of the total ankle path length. In these examples, a full recovery is made back to the unperturbed mean, indicating the placing phase of these wiping movements is stable about the final midline position.

The magnitude of the applied perturbation ranged from 0.35 to 1.50 N and the duration ranged from 25 to 75 ms. For the example in Figure 4-1, the applied force was 1.25 N for 25 ms. These brief and relatively large perturbations rapidly changed

the mechanical state of the limb (position and velocity) in a manner analogous to changing initial conditions of a dynamic system with a delta function input. Upon perturbation offset, acceleration- and possibly velocity-dependent inertial forces carry the multijoint hindlimb further from the unperturbed path before compensating forces dominant the motion. No gravitational forces are involved since these movements are restricted to the horizontal plane.

4.1.2 Perturbed position

The effect of the perturbation on the position of hindlimb movements was analyzed to assess the statistical and functional significance of the perturbation. The initial displacement due to the perturbation was defined as the maximum displacement from the mean unperturbed ankle path, calculated along directions perpendicular to the path, in the 100 ms following perturbation offset.

Afferented hindlimb wipe

The cumulative distribution of initial displacement for all trials of all eight frogs for the afferented hindlimb wipe condition is shown in Figure 4-2A (109 O unperturbed trials, 85 CK perturbed trials, 105 CCK perturbed trials). The difference between perturbation group (O, CK, CCK) distributions is highly statistically significant (pairwise Kolmogorov-Smirnov tests, all p << 0.001) indicating the perturbations exceeded the variability of the movement. The upper abscissa gives a measure of the functional significance of the perturbed position, relating it to the fraction of the total unperturbed path length averaged across all eight frogs. The clockwise (CK) perturbations tended to cause less of a displacement than counterclockwise (CCK) trials. This may be due to anisotropy of the apparent ankle impedance, a phenomenon associated with multijoint limbs [17].

The initial displacement in joint coordinates (calculated using the same method as for ankle coordinates) is given for two of the eight frogs (Fig. 4-2B): one (Frog 21) that had small to moderate ankle initial displacement and one (Frog 19) that had



Figure 4-2: Initial displacements for hindlimb wipe movements

large ankle initial displacement. For both frogs, the perturbations caused a larger change in hip angle than knee angle. For frog 19, the change in hip and knee angle was the same sign: both extension (positive) for CCK perturbations and both flexion (negative) for CK perturbations. For frog 21, the change in hip and knee angle differed in sign. The fraction of joint angle change to mean unperturbed joint movement, a measure of the functional significance of the perturbed joint position, is indicated on the right ordinate.

Hindlimb withdrawal

The cumulative distribution of initial displacement for all trials of all six frogs for the hindlimb withdrawal condition is shown in Figure 4-3A (101 unperturbed trials, 72 CK perturbed trials). As with the wiping movements, the



Figure 4-3: Initial displacements for hindlimb withdrawal movements

difference between perturbation group distributions was highly statistically significant (pairwise Kolmogorov-Smirnov tests, all $p \ll 0.001$), again indicating the perturbations exceeded the variability of the movement. Overall, the initial displacements in the withdrawal (middle 50th percentile range for CCK perturbed trials: 2.07 mm to 4.99 mm) were smaller than for the wipe (middle 50th percentile range for CCK perturbed trials: 4.02 mm to 8.21 mm). This difference is at least partially a result of smaller force perturbation magnitudes applied to withdrawal movements (mean = 0.53 N) as compared to wiping movements (mean = 0.62 N). Like the wiping movements, there was some tendency for CK perturbations to cause less displacement than CCK perturbations, suggesting a directionality to the ankle impedance function.

The initial displacement in joint coordinates for withdrawal movements is given for two of the six frogs (Fig. 4-3B). The two frogs are representative of the total ankle perturbation position range. In contrast with the wiping movements, perturbations to withdrawal movements caused approximately the same amount of deviation in knee angle and hip angle and the sign of the hip and knee angle changes differed.

Thus the perturbations to both hindlimb wipe and withdrawal behaviors caused the path of the hindlimb to significantly deviate from unperturbed paths. To extend this analysis of the effect of applied perturbations on hindlimb kinematics, changes in hindlimb velocity are considered in the next section.

4.1.3 Perturbed velocity

In looking at perturbation-induced changes in hindlimb velocity, the particular focus was to determine the likelihood of muscle spindle response. Studies in humans have found that as change in joint velocity following a perturbation increases, reflexive response to the perturbation is suppressed [41]. Therefore rather than present velocity changes in ankle coordinates, to facilitate an estimate of spindle response the velocities are expressed in joint and muscle (ST,BF) coordinates. Perturbation induced velocity changes were analyzed for the same two pairs of frogs used to examine perturbed joint positions in the previous section. Velocity profiles for each trial were aligned by perturbation onset and averaged for the first 200 ms following onset. Within this time window the velocity profiles were relatively consistent across trials and exhibited the major velocity changes. The transformation from joint to muscle coordinates used muscle-specific, configuration-dependent moment arms found for Rana pipiens by Kargo and Rome [23], scaled up by ratio of tibiofibula length for the bullfrogs used in this study.

Perturbed velocities for both wipe (Fig. 4-4A) and withdrawal (Fig. 4-4B) movements exhibited a tendency to oscillate about the unperturbed velocity, rather than returning monophasically. Changes in muscle velocity following perturbation did not exceed 20 mm/s. This is well within the dynamic range of spindle response and, according to results from Ottoson [34], would be expected to cause a spike frequency change on the order of 20 impulses per second over basal, unperturbed levels. Note



Figure 4-4: Perturbed velocities for hindlimb wipe and withdrawal movements

that in Figure 4-4 muscle shortening velocities are positive while lengthening is negative and perturbation groups (O, CW, CCW) are indicated by the same colors as in Figures 4-2 and 4-3.

The perturbed position and velocity achieved in both hindlimb wipe and withdrawal movements substantially exceed the variability in the unperturbed movement, likely making recovery from the perturbation behaviorally relevant and allowing an analysis of any such recovery. Also, the perturbation caused kinematic changes to which muscle spindles should be capable of responding. The rest of this chapter presents a kinematic analysis of the perturbation recovery.

4.2 Final position stability

4.2.1 Hindlimb wipe

Stability about the final hindlimb position was analyzed across the eight afferented wiping frogs by examining the cumulative distributions of unperturbed and perturbed trial final position distances from unperturbed mean final position (Fig. 4-5). Pairwise Kolmogorov-Smirnov tests found no pair of distributions that significantly differed (p > 0.1 for all three tests), indicating that the perturbed trial final positions where indistinguishable from unperturbed trial final positions. The magnitude of recovery



Figure 4-5: Final position summary for hindlimb wipe movements

can be seen by comparing the cumulative distributions in Figure 4-5 to those in Figure 4-2A. As a within session (8 frogs, 9 sessions) test of final position stability, a one-way multiple analysis of variance (MANOVA) on the final positions for the three groups of trials found six out of the nine sessions showed no significant difference in final position at the $\alpha = 0.05$ level (using the Wilks' lambda test statistic which has an exact F distribution for two dependent variables and three groups, see Johnson and Wichern [19]). All three sessions with a significant main effect had a significant pairwise effect between unperturbed trials and CCK perturbed trials at the $\alpha = 0.05$ level (using Bonferroni method of familywise significance adjustment for multiple tests).

4.2.2 Hindlimb withdrawal

Stability about the final position in withdrawal movements was comparable to that of the placing phase of wiping movements. The final position was defined as the point of first contact with the body. The cumulative distributions of unperturbed and perturbed trial final position distances from unperturbed mean final position for all six withdrawal frogs is shown in Figure 4-6. All pairwise Kolmogorov-Smirnov



Figure 4-6: Final position summary for hindlimb withdrawal movements

tests were not significant (p > 0.1 for all three tests). The magnitude of recovery can be seen by comparing the cumulative distributions in Figure 4-6 to those in Figure 4-3A. Within session (6 frogs, 10 sessions) MANOVA tests found seven of 10 sessions were not significant at the $\alpha = 0.05$ level. All three sessions with a significant main effect had a significant pairwise effect between unperturbed trials and CK perturbed trials at the $\alpha = 0.05$ level with Bonferroni correction.

The results on final position stability indicate that withdrawal movements and the placing phase of wiping movements are generally stable about their respective final positions. The ability of spinal motor systems to stabilize movement is not perfect, however, as evident in the within session analyses. Before exploring the mechanism responsible for stabilizing movement, a stronger definition of stability is tested: one which considers the full mechanical state rather than just final position.

4.3 Trajectory stability

In the final position stability analysis presented above, the point of midline contact for the wipe and of body contact for the withdrawal were taken as equilibrium points for the closed-loop system dynamics. These equilibrium points were in general found to be stable. For the following analysis, stability is defined not with respect to an equilibrium point, but rather in terms of the tendency for any two perturbed system trajectories to converge towards each other.

4.3.1 Contraction analysis

Contraction theory provides a necessary and sufficient condition for determining regions of the state space of a general deterministic nonlinear system, $\dot{\mathbf{x}} = \mathbf{f}(\mathbf{x}, t)$ for which all trajectories exponentially converge to each other [29]. Systems with this property, termed contracting or incrementally stable [1] systems, have a number of advantages, such as stability in combination and open-loop observability, that may be particularly useful for biological control [39]. This analysis was therefore applied to the frog wiping and withdrawal data collected in this study to determine if the spinal motor systems of the frog may have these advantages. To adapt the usual, non-experimental, application of contraction theory to the present data, kinematic equivalents of the theorems from contraction theory (as stated below) were used. This permitted an analysis based directly on the measured kinematics, independent of an explicit system model, but also restricted the conclusions to apply only to portions of the state space containing the observed trajectories.

Convergence in ankle and joint coordinates

If the system dynamics, $\mathbf{f}(\mathbf{x}, t)$, are known, the sufficient condition for contraction is simply that the system Jacobian, $\partial \mathbf{f}/\partial \mathbf{x}$, is uniformly negative definite for all \mathbf{x} and t. If the system dynamics are not known, as is assumed for the model independent analysis of the frog hindlimb performed here, an equivalent sufficient condition for contraction is that the rate of change of squared distance between every pair of system trajectories, $d/dt(\Delta \mathbf{x}^T \Delta \mathbf{x})$, is less than zero for all t, where $\Delta \mathbf{x}$ is the difference between the full state vector (a 4 x 1 vector for the planar motion of the hindlimb and assuming the dominant dynamics are second order) of two trajectories. In other words, if the squared distance between each pair of system trajectories is monotonically decreasing for all time, the system is contracting.



Figure 4-7: Contraction analysis in ankle coordinates

Within each frog, $\Delta \mathbf{x}^T \Delta \mathbf{x}$ and its rate of change were computed for each pair of CK and CCK perturbed trials from perturbation offset to the final position, as previously defined. Examples are shown in Figure 4-7 for two frogs, where the mean time course (\pm one standard deviation) of each quantity, computed using ankle coordinates, is indicated. In joint coordinates the results were nearly identical. The average squared relative distance between pairs of CK and CCK perturbed trajectories did not monotonically decrease in any of the frogs analyzed. These results indicate that the sufficient condition could not be met by these movements, and thus no conclusion about incremental stability can be drawn from this analysis.

Convergence in alternate coordinates

The above results show that the trajectories produced by spinalized frogs are not contracting using the basic analysis described there. However, although the conditions described above are sufficient to show contraction, they are not necessary, and in fact the system in question might still be contracting using a slightly different and more general analysis. In such generalized contraction analysis, the necessary condition for contraction is that there exist some uniformly positive definite metric with respect to which the system can be shown to be contracting [29]. Note that for the contraction condition to be necessary and sufficient it must allow for nonautonomous metrics. This metric acts to transform the system dynamics such that the contraction condition described in the previous section is now satisfied by the transformed system. If such a metric can be found, then the system can be considered to be contracting.

The difficulty in this more general analysis, however, is in finding an appropriate metric. In cases where the system dynamics are known analytically, one can in some cases make educated guesses as to possible forms of suitable metrics. In the present case of examining the movements produced by the frog hindlimb, however, in which we do not have a suitable model of the dynamical system, it is not clear how to choose a suitable metric.

We have therefore taken a different but equivalent approach to this issue [40]. It can be shown that finding a metric in which system is contracting is equivalent to finding some stable transfer function $1/(a_0 + a_1s + \ldots + a_ms^m)$ with positive impulse response such that

$$a_0 \Delta \mathbf{x}^T \Delta \mathbf{x} + a_1 \frac{d}{dt} (\Delta \mathbf{x}^T \Delta \mathbf{x}) + \ldots + a_m \frac{d^m}{dt^m} (\Delta \mathbf{x}^T \Delta \mathbf{x}) \le 0.$$

If such a m^{th} -order transfer function can be found, then this condition implies the system is contracting. The advantage of this expression of the generalized contraction condition is that it is much easier to find possible stable transfer functions and this search can therefore be performed using standard optimization routines rather than trial and error approaches of choosing different metrics. We do however simplify the analysis by restricting the transfer function to be time-invariant, corresponding to an autonomous metric.

Using this analysis, we examined different orders of stable transfer functions and assessed whether the above contraction condition could be met. We used a constrained nonlinear optimization routine, which, starting from some initial values of the transfer function, attempted to find new values which minimized the maximum value of the above condition, subject to the constraint that the transfer function be stable and have a positive impulse response function. Several objective functions other than maximum value were used, including the sum of positive values and the number of positive values, yielding similar results. The initial values of the transfer function were



Figure 4-8: Contraction analysis in alternate coordinates

identified by using an exhaustive search for the transfer function poles (searching an evenly spaced logarithmic grid from 0.01 to 4000) which led to the best contraction conditions so that minima found by the optimization would be close to the global minimum for this range of pole values. The results of this analysis using 1^{st} , 4^{th} , and 8^{th} order transfer functions are shown in Figure 4-8A (for frog 19 wiping movements, mean shown). As can be seen in the figure, including higher order terms to the transfer function substantially improved this condition, minimizing the positive portion of these functions. Figure 4-8B shows how the maximum positive value decreased with higher order terms. However, even with the higher order terms included in this analysis, we were not able to find a transfer function for which these functions were uniformly negative. This analysis was repeated using the 20 best initial conditions from the exhaustive search described above and using data from other frogs and in no case was a transfer function found which satisfied the contracting condition for all times. Based on this analysis, therefore, we cannot conclude that this system is contracting.

Chapter 5

Stability control strategy

The previous chapter established that movements produced by the frog spinal cord are able to compensate for phasic perturbations in order to achieve the unperturbed final position. We assume that motor systems in the frog spinal cord are stabilizing these movements, since forces arising from reaching the joint limits or contacting the environment were avoided in these experiments. Spinal motor systems may stabilize movement passively through the viscoelastic properties of muscles or actively through reflexes. To distinguish between these two possibilities, two approaches were taken. First, the stability of hindlimb wipes in the deafferented condition was examined. Second, EMGs from afferented movements were analyzed for perturbation-induced changes.

5.1 Kinematic evidence from deafferented frogs

To determine the ability of intrinsic muscle properties to stabilize movement, eight frogs were unilaterally deafferented and tested with the same perturbation paradigm that was used with the afferented frogs. In the deafferented condition, hindlimb wiping movements could be evoked with cutaneous stimulation as before. However hindlimb withdrawals could not be evoked with cutaneous stimulation, since only the leg contralateral to the deafferented side could sense a cutaneous stimulus. Thus only wiping movements were examined in this condition. The effect of the applied perturbation on the hindlimb path for all eight deafferented frogs is summarized in Figure 5-1A (123 O unperturbed trials, 95 CCK perturbed trials, 94 CK perturbed trials). The perturbations applied to the deafferented wiping movements (middle 50th percentile range for CCK perturbed trials: 5.00 mm to 11.68 mm) were generally larger than those applied to the afferented wiping movements (middle 50th percentile range for CCK perturbed trials: 4.02 mm to 8.21 mm).



Figure 5-1: Perturbation and stability summary for deafferented movements

The deafferented frogs were able to substantially compensate for the perturbations. The perturbed trials, with 95% confidence interval for the unperturbed mean, for two deafferented frogs are shown in Figure 5-2 in ankle coordinates. The squares indicate the starting position of the movements. To quantify the stability about the



Figure 5-2: Example perturbed ankle paths from deafferented frogs

final position, the distributions of final displacement from unperturbed mean for all trials were computed (Fig. 5-1B). While a large recovery was seen (comparing Fig. 5-1A to Fig. 5-1B), pairwise Kolmogorov-Smirnov test found both the unperturbed and CCK perturbed distributions and the unperturbed and CK perturbed distributions to be significantly different (p = 0.0112 and p = 0.0053, respectively). Furthermore, within session MANOVA tests found four of nine sessions had significantly ($\alpha = 0.05$ level) different final positions by perturbation group. Each of the four sessions with significant main effects had a significant pairwise effect between the unperturbed and CK perturbed group at the $\alpha = 0.05$ level with Bonferroni correction. These statistical tests suggest that afferent compensation normally plays some role in stabilizing the frog hindlimb.

Further support for the role of afferents in this system is seen in Figure 5-3. All CK and CCK perturbation groups for wiping frogs with an average initial displacement less that 10 mm are plotted against their corresponding average final displacement. The requirement on initial displacement size ensures that any correlation of these variables observed in the afferented wipes or in the deafferented wipes is found under



Figure 5-3: Relationship between initial and final displacements for wiping movements

comparable perturbation conditions. No significant correlation (p = 0.286) between initial and final displacement is seen for afferented frogs. However these variables are significantly correlated (p < 0.001) in deafferented frogs, with initial displacements accounting for 62% of the variance in observed final displacements. First, this result suggests that intrinsic muscle properties provide insufficient impedance to fully compensate for the perturbations used in these experiments. Second, afferent feedback likely plays some role in making the final displacement independent of the initial displacement. In light of this result, we next look specifically for any afferent involvement in the compensatory response by examining modulation in perturbed trial EMGs.

5.2 EMG evidence from afferented frogs

Differences in EMG timing, magnitude, or duration between unperturbed and perturbed trials would provide evidence for afferent involvement in the perturbation rejection. Therefore, a statistical analysis of changes in these parameters was carried out for the afferented hindlimb wipe and withdrawal movements. The relevant EMG latencies to consider for wiping movements are the same as those considered in Chapter 3: the four onset-to-onset latencies between ST/BF and RI/SM. In chapter 3, the ST-RI latency was found to be significantly correlated with nominal movement variability and to be modified by afferent feedback. The three other latencies have been found to be modified by afferent feedback in the wiping behavior by other studies [21]. EMG magnitude was calculated for these four muscles for each wiping trial (just ST and BF for withdrawal trials) by integrating the perturbation-aligned, rectified, and filtered EMG from perturbation onset to final position. This definition of EMG magnitude maximized the likelihood of observing an afferent effect associated with the perturbation. The duration (onset to offset) was also calculated for ST, BF, RI, and SM for each wiping trial (just ST and BF for withdrawal trials). EMG durations have been found previously to be modulated by muscle afferents in nominal wiping movements [21]. Within each afferented frog, each EMG parameter was used as the dependent variable in a three-level (perturbation group: O, CK, CCK), one-way nonparametric analysis of variance (Kruskal-Wallis ANOVA).

Out of all 132 tests, only eight reached significance at the $\alpha = 0.05$ level. Of these eight tests, ST magnitude and BF magnitude were the only EMG parameters with a frequency of significance greater than chance. A combined 6 out of 30 ST and BF tests reached significance. This result agrees with the previous analysis in deafferented frogs, namely that afferent feedback is utilized in the perturbation response. This result also makes sense with respect to the behavior. ST and BF are generally the most active muscles at the onset of movement, both in wipe and withdrawal, and at the time of the perturbation. Therefore the inputs to these muscles are the most likely to be modified by homonymous, and possibly heteronymous, feedback following the perturbation. However of these six ST and BF tests that reached significance, only two showed magnitude differences that appeared to be functionally significant. The other four have mean changes in magnitude of less than 20% from the mean unperturbed value. An example of a large and a small perturbation-induced EMG change, both of which were statistically significant, is shown in Figure 5-4. Frog 19 had a statistically significant change in ST and frog 21 had a statistically significant change in BF following the perturbation. The change in BF magnitude in frog 21, however, was small: an 18% change for CK perturbations and a 14% change for CCK perturbations over the mean unperturbed magnitude. Therefore upon closer analy-



Figure 5-4: Mean EMGs for two afferented wiping frogs

sis, the EMG data from afferented frogs does not overwhelmingly support afferent involvement in movement stabilization.

The deafferented wipe data indicates that both intrinsic muscle properties and afferent feedback are involved in controlling hindlimb movement stability to some degree. However, the relatively large contribution of intrinsic impedance to perturbation compensation, as seen in Figure 5-2, may suggest why only small perturbationinduced EMG changes were found. A large reflexive action does not appear to be necessary for movement stabilization based on these data.

Chapter 6

Discussion and conclusion

This thesis analyzed the mechanical stability of hindlimb wipe and withdrawal movements produced by spinalized frogs. The movements were found to generally be stable about the final position. Furthermore, the data suggest that the movements were stabilized by a combination of intrinsic and reflexive mechanisms.

Movement stability analyses

The stability analyses were performed using the kinematic data collected for hindlimb wipe and withdrawal movements. The first analysis compared the final position of the perturbed trials to the unperturbed trials to assess the stability about this presumed equilibrium point. This simple stability property is sometimes referred to as equifinality in motor control literature and is a necessary condition for the equilibrium point hypothesis [37]. For both afferent wipe and withdrawal movements, the distributions of final positions did not significantly or functionally differ between perturbation groups. Thus the property of equifinality was verified for these movements. However a few exceptions to this conclusion did occur. These exceptions could either represent excessive noise in our estimate of the unperturbed final position or reflect real movement instability, as has been seen in some human arm movements without voluntary corrections [14].

The second stability analysis utilized contraction theory to test the spinal motor system for a stronger version of stability, one in which all system trajectories exponentially converge toward each other. One of the significant features of this property is that it allows nonlinear systems to be combined stably. This property may underlie the successful parallel combination of movement primitives for the construction of stable movement [39]. However, the wipe and withdrawal movements did not strictly meet the conditions for a contracting system. One complicating factor is that contraction theory was developed for deterministic systems, whereas biological systems are certainly stochastic. Therefore meeting a strict deterministic criterion for contraction is likely not necessary for meaningfully characterizing a biological system as contracting. Also, the analysis used is this thesis was not the most general analysis that could be performed for assessing whether a system is contracting. We restricted our analysis to looking for autonomous metrics in which the system was contracting, via the equivalent approach described in Section 4.3.1. However nonautonomous metrics must be considered to have a necessary and sufficient contracting condition. Further work could extend the optimization routine described in Section 4.3.1 to allow for nonautonomous metrics. Another consideration is that the dynamics of the robot that was attached to the hindlimb may have prevented the coupled system from being contracting. The mass of the robot arm was similar to the mass of the bullfrog hindlimb. Therefore one might expect significant configuration-dependent forces at the interface point on the hindlimb during movement. Future work could estimate the influence of these forces, perhaps using a model-based controller to make the apparent robot endpoint inertia small and isotropic, on the stability results summarized above.

Analyses of the stability controller

Two approaches were taken to identify the degree to which muscle viscoelasticity and reflexes were involved in stabilizing the movements. The first approach used a surgical means, deafferentiation, to differentiate between the two mechanisms. This has been a classic approach to the general question of how intrinsic and reflexive mechanisms contribute to limb impedance [33] [15]. After deafferentiation, the hindlimb movements were found to substantially compensate for the perturbations, but failed to achieve the same level of compensation as seen with afferents intact. Furthermore, in the deafferented condition, the final displacement was roughly proportional to the initial displacement caused by the perturbation. This proportional relationship was not true of afferented movements. The relationship between initial and final displacements in the deafferented frog suggests that the intrinsic muscle properties provide relatively low-gain mechanical feedback that cannot fully compensate for the perturbations applied in these experiments. Both of these findings suggest that afferent feedback is likely to play some role in stabilizing the hindlimb movements.

In the second approach, to more directly explore reflex action, we looked for reflexive EMG modifications following the perturbation of afferented movements. Note that the method of delivery of perturbations to the hindlimb, via a bone screw, probably prevented much of a cutaneous response. Thus any afferent feedback used to stabilize the limb was dominated by proprioceptive information and specifically lacked the cutaneous information a more natural perturbation, one that contacts the skin, would elicit. Evidence from a previous study suggests that cutaneous feedback is not used to regulate the placing phase of wiping behaviors [21]. Instances of postperturbation changes in the EMG driven by this proprioceptive information were very few. This was not expected given the conclusion from the deafferented frogs that afferents play a role in limb stabilization. However in the previous analysis we found that intrinsic muscle properties were large enough to substantially compensate for the perturbations (Fig. 5-2). Therefore only small modifications to muscle activations were likely needed to complement this intrinsic compensation. These small modifications may not have been distinguishable from noise given the limited resolution of the EMG recordings. We might expect that with larger perturbations of the afferented movement, more significant EMG changes would be found. Future experiments could focus on exposing afferented movements to larger perturbations, within the limits of reaching workspace boundaries, to clarify the role of afferents in the stabilization of this system.

Relation to intraspinal stimulation experiments

The stability results of these experiments extend the observations of convergent isometric force patterns, or primitives, evoked from intraspinal stimulation in the frog [12] [32]. In these isometric experiments, the full position- and velocity-dependence of the neuromuscular force could not be characterized. Furthermore, sufficient data exists to suggest that in some biological systems, perhaps for example humans, movements can be made for which intrinsic and reflexive action do not sufficiently stabilize the limb [37] [14]. Therefore, the presence of convergent neuromuscular forces under isometric conditions cannot be used to conclude that movement will be dynamically stable. Assuming the cutaneously-evoked movements of this study were produced by accessing similar spinal motor circuits as tested with intraspinal stimulation, the results of this thesis show that the spinal primitives, at least when acting in combination, can indeed stabilize movement. Validating the above assumption, evidence connecting the spinal force patterns of intraspinal stimulation to cutaneously-evoked behaviors has been reported in a study showing summation of primitives during corrective responses in cutaneously-evoked wipes [22].

Future directions

Exploring the mechanical properties of motor behaviors provides insight into how the nervous system may control movement, as seen in theories such as equilibrium point control and impedance control. An experimental approach to characterizing these properties, as used in this thesis, ultimately provides the test for such theories. However access to and manipulation of many relevant mechanical variables, such as muscle length, velocity, force, and moment arms, is restricted in animal models. With the development of detailed biomechanical models, such as the ones recently developed by Kargo and colleagues [23] [24], mechanical properties can be explored in much greater detail without sacrificing much of their true character. Using a detailed musculoskele-tal model may be a next step in testing the spinal primitive-based equilibrium point controller. Specifically, one could test the impedance magnitude requirements of indi-

vidual primitives, mechanical advantages of certain combinations of muscles usually found to activate synchronously (e.g., ST and BF or RI and SM), and mechanical properties important for activating primitives in parallel to produce movement. Using these methods, in combination with the experimental methods described in this thesis, should provide important insight into how the mechanical properties shape neural control and how neural control shapes the mechanical properties of motor behaviors.

Bibliography

- Angeli D. (2002) A Lyapunov approach to incremental stability properties. *IEEE Trans. Automat. Contr.* 47:410-421.
- [2] Bernstein N. (1967) The Coordination and Regulation of Movements, New York: Pergamon Press.
- Bennett D.J. (1993) Torques generated at the human elbow joint in response to constant position errors imposed during voluntary movements. Exp. Brain. Res. 95:488-498.
- [4] Bizzi E., Accornero N., Chapple W., and Hogan N. (1982) Arm trajectory formation in monkeys. Exp. Brain Res. 46:139-143.
- [5] Bizzi E., Mussa-Ivaldi F.A., and Giszter S. (1991) Computations underlying the execution of movement: a biological perspective. *Science* 253:287-291.
- [6] Bizzi E., Hogan N., Mussa-Ivaldi F.A., and Giszter S. (1992) Does the nervous system use equilibrium-point control to guide single and multiple joint movements? Behav. Brain Sci. 15:603-613.
- [7] Burdet E., Osu R., Franklin D.W., Milner T.E., and Kawato M. (2001) The central nervous system stabilizes unstable dynamics by learning optimal impedance. *Nature* 414:446-449.
- [8] Dolan J.M., Friedman M.B., and Nagurka M.L. (1993) Dynamic and loaded impedance components in the maintenance of human arm posture. *IEEE Trans.* Sys. Man Cybern. 23:698-709.

- [9] Feldman A.G. (1966) Functional tuning of the nervous system during control of movement or maintenance of a steady posture. III. Mechanographic analysis of the execution by man of the simplest motor tasks. *Biophysics* 11:766-775.
- [10] Feldman A.G. (1986) Once more on the equilibrium-point hypothesis (λ model) for motor control. J. Mot. Behav. 18:17-54.
- [11] Giszter S.F., McIntyre J., and Bizzi E. (1989) Kinematic strategies and sensorimotor transformations in the wiping movements of frogs. J. Neurophysiol. 62:750-767.
- [12] Giszter S.F., Mussa-Ivaldi F.A., and Bizzi E. (1993) Convergent force fields organized in the frog's spinal cord. J. Neurosci. 13:467-491.
- [13] Gomi H. and Kawato M. (1997) Human arm stiffness and equilibrium-point trajectory during multijoint movement. *Biol. Cybern.* 76:163-171.
- [14] Hinder M.R. and Milner T.E. (2003) The case for an internal dynamics model versus equilibrium point control in movement. J. Physiol. in press.
- [15] Hoffer J.A. and Andreassen S. (1981) Regulation of soleus muscle stiffness in premammillary cats: intrinsic and reflex components. J. Neurophysiol. 45:267-285.
- [16] Hogan N. (1984) An organizing principle for a class of voluntary movements. J. Neurosci. 4:2745-2754.
- [17] Hogan N. (1985) The mechanics of multi-joint posture and movement control. Biol. Cybern. 52:315-331.
- [18] Hogan N. (1985) Impedance control: an approach to manipulation. ASME J. Dynamic Syst. Meas. Contr. 107:1-24.
- [19] Johnson R.A. and Wichern D.W. (1992) Applied Multivariate Statistical Analysis (3rd ed.), Englewood Cliffs, NJ: Prentice Hall.

- [20] Joyce G., Rack P.M.H., and Westbury D.R. (1969) The mechanical properties of cat soleus muscle during controlled lengthening and shortening movements. J. Physiol. 204:461-474.
- [21] Kargo W.J. and Giszter S.F. (2000) Afferent roles in hindlimb wipe-reflex trajectories: free-limb kinematics and motor patterns. J. Neurophysiol. 83:1480-1501.
- [22] Kargo W.J. and Giszter S.F. (2000) Rapid correction of aimed movements by summation of force-field primitives. J. Neurosci. 20:409-426.
- [23] Kargo W.J. and Rome L.C. (2002) Functional morphology of proximal hindlimb muscles in the frog Rana pipiens. J. Exp. Biol. 205:1987-2004.
- [24] Kargo W.J., Nelson F., and Rome L.C. (2002) Jumping in frogs: assessing the design of the skeletal system by anatomically realistic modeling and forward dynamic simulation. J. Exp. Biol. 205:1683-1702.
- [25] Kubow T.M. and Full R.J. (1999) The role of the mechanical system in control: a hypothesis of self-stabilization in hexapedal runners. *Phil. Trans. R. Soc. Lond.* B 354:849-861.
- [26] Lacquaniti F., Carrozzo M., and Borghese N.A. (1993) Time-varying mechanical behavior of multijointed arm in man. J. Neurophysiol. 69:1443-1464.
- [27] Lemay M.A. and Grill W.M. (2000) Endpoint force patterns evoked by intraspinal stimulation - ipsilateral and contralateral responses. Proc. 22nd EMBS Int. Conf. 918-919.
- [28] Liddell E.G.T. and Sherrington C. (1924) Myotactic reflexes. Proc. Roy. Soc. Lond. B 96:212-242.
- [29] Lohmiller W. and Slotine J-.J.E (1998) On contraction analysis for non-linear systems. Automatica 34:683-696.
- [30] McIntyre J. and Bizzi E. (1993) Servo hypotheses for the biological control of movement. J. Mot. Behav. 25:193-202.

- [31] Mussa-Ivaldi F.A., Hogan N., and Bizzi E. (1985) Neural, mechanical, and geometrical factors subserving arm posture in humans. J. Neurosci 5:2732-2743.
- [32] Mussa-Ivaldi F.A., Giszter S.F., and Bizzi E. (1994) Linear combinations of primitives in vertebrate motor control. Proc. Natl. Acad. Sci. USA 91:7534-7538.
- [33] Nichols T.R. and Houk J.C. (1976) Improvement in linearity and regulation of stiffness that results from actions of stretch reflex. J. Neurophysiol. 39:119-142.
- [34] Ottoson D. (1976) Morphology and physiology of muscle spindles. In Frog Neurobiology (ed. R. Llinas and W. Precht), pp. 643-675. New York: Springer-Verlag.
- [35] Perreault E.J., Kirsch R.F., and Acosta A.M. (1999) Multiple-input, multipleoutput system identification for characterization of limb stiffness dynamics. *Biol. Cybern.* 80:327-337.
- [36] Perreault E.J., Kirsch R.F., and Crago P.E. (2002) Voluntary control of static endpoint stiffness during force regulation tasks. J. Neurophysiol. 87:2808-2816.
- [37] Popescu F.C. and Rymer W.Z. (2000) End points of planar reaching movements are disrupted by small force pulses: an evaluation of the hypothesis of equifinality. J. Neurophysiol. 84:2670-2679.
- [38] Schotland J.L. and Rymer W.Z. (1993) Wipe and flexion reflexes of the frog. I. Kinematics and EMG patterns. J. Neurophysiol. 69:1725-1735.
- [39] Slotine J-.J.E. and Lohmiller W. (2001) Modularity, evolution, and the binding problem: a view from stability theory. *Neural Netw.* 14:137-145.
- [40] Slotine J-.J.E. personal communication.
- [41] Stein R.B. and Kearney R.E. (1995) Nonlinear behavior of muscle reflexes at the human ankle joint. J. Neurophysiol. 73:65-72.
- [42] Tsuji T., Morasso P.G., Goto K., and Ito K. (1995) Human hand impedance characteristics during maintained posture. *Biol. Cybern.* 72:475-485.

[43] Won J. and Hogan N. (1995) Stability properties of human reaching movements. Exp. Brain Res. 107:125-136.